Author's response to reviews

Title: Reliability and validity study of Persian modified version of MUSIC (musculoskeletal intervention center) - Norrtalje questionnaire

Authors:

Akbar Alipour (akbar.alipour@ki.se)
Mostafa Ghaffari (mostafa.ghaffari@ki.se)
Irene Jensen (irene.jensen@ki.se)
Batoul Shariati (batoul_shariati@yahoo.com)
Eva Vingard (eva.vingard@medsci.uu.se)

Version: 5 Date: 4 August 2007

Author's response to reviews: see over
Dear Editor in Chief and Reviewers in BMC

Please accept our thanks for accepting our manuscript for publication.
We also appreciate your final comments for last version of manuscript; we have revised our manuscript accordingly:
1. We removed the authors' qualifications/titles and wrote their list as they are in submission system
2. Key words were removed.
3. We changed former titles to background and method
4. We put tables after reference
5. Figure were changed based on your journal format
6. We again checked references and their citing format (in square brackets) in text.

Sincerely Yours,

Akbar Alipour                           Eva Vingard                    Iren Jensen
MD  PhD student                      MD-PhD                          PhD
Batoul Shariati                          Mostafa Ghaffari
MD-PhD                                   MD-MPH

Correspondence: Akbar Alipour- MD
Address: Uppsala University / Department of Medical Sciences – Occupational and Environmental Medicine. Ulleräkersvägen 38-40, SE-751 85 Uppsala-SWEDEN
Telephone number: 0046-18-611 97 44
Fax number: 0046-18-51 99 76
Email: akbar.alipour@medsci.uu.se